Author's response to reviews

Title: 'Relief of oppression': An organizing principle for researchers' obligations to participants in observational studies in the developing world

Authors:

James V Lavery (jim.lavery@utoronto.ca)
Sunita VS Bandewar (sunita.bandewar@utoronto.ca)
Joshua Kimani (jkimani@csrtkenya.org)
Ross EG Upshur (ross.upshur@utoronto.ca)
Frances A Plummer (Frank_Plummer@phac-aspc.gc.ca)
Peter A Singer (peter.singer@mrcglobal.org)

Version: 4 Date: 18 June 2010

Author's response to reviews:

Final revisions:

1. Changed title to all title case
2. Deleted date on title page
6. Replaced previous reference #6 with a more comprehensive reference from UNAIDS